Author's response to reviews

Title: The publics' understanding of daily caloric recommendations and their perceptions of calorie posting in chain restaurants

Authors:

Sara N Bleich (sbleich@jhsph.edu)
Keshia M Pollack (kpollack@jhsph.edu)

Version: 6 Date: 5 March 2010

Author's response to reviews: see over
March 5, 2010

Melissa Norton, MD
Editor-in-Chief, BMC Public Health
BioMed Central Ltd
Floor 6
236 Gray's Inn Road
London WC1X 8HL
United Kingdom

Dear Dr. Norton,

Please find the attached updated manuscript entitled “The publics' understanding of daily caloric recommendations and their perceptions of calorie posting in chain restaurants.” We are thrilled that the manuscript has been accepted. Minor revisions have been made.

Thank you.

Sincerely,

Sara N. Bleich
Assistant Professor
Department of Health Policy and Management
Johns Hopkins School of Public Health
624 N. Broadway, Room 451
Baltimore, MD 21205